Author's response to reviews

Title: Determination of quality of life in adolescents with idiopathic scoliosis subjected to conservative treatment.

Authors:

Angelo G Aulisa (agaulisa@tiscali.it)
Vincenzo Guzzanti (vguzzanti@yahoo.it)
Carlo Perisano (carloperisano@hotmail.it)
Emanuele Marzetti (emarzetti@aging.ufl.edu)
Alessandro Specchia (alessandro_specchia@rm.unicatt.it)
Marco Giordano (m.giordano@opbg.net)
Lorenzo Aulisa (aulisa@libero.it)

Version: 5 Date: 26 September 2010

Author's response to reviews:

Major revisions:
-the tables are in the text after the references
-the references are now cited in numerical order (pag 14)
-Figure 4 are cited in the text (pag 11)